Author's response to reviews

Title: Hemicraniectomy After Middle cerebral artery infarction with Life-threatening Edema Trial (HAMLET). Protocol for a randomised controlled trial of decompressive surgery in space-occupying hemispheric infarction

Authors:

Jeannette Hofmeijer (j.hofmeijer@umcutrecht.nl)
G Johan Amelink (g.j.amelink@umcutrecht.nl)
Ale Algra (a.algra@umcutrecht.nl)
Jan van Gijn (j.vangijn@umcutrecht.nl)
Malcolm R Macleod (malcolm@apoptosis.freeserve.co.uk)
L Jaap Kappelle (l.kappelle@umcutrecht.nl)
H Bart van der Worp (h.b.vanderworp@umcutrecht.nl)

Version: 2 Date: 4 September 2006

Author's response to reviews:

Utrecht 04.09.06

Subject: Submission revised MS 1797211743111320

Dear Editor,

Please find enclosed the revised manuscript "Hemicraniectomy After Middle cerebral artery infarction with Life-threatening Edema Trial (HAMLET). Protocol for a randomised controlled trial of decompressive surgery in space-occupying hemispheric infarction".

Editorial changes:

Please remove the ISRCTN number from the title. It is fine in the abstract. Done.

Formatting changes:

Title: Please remove the full stop from the end of the title. Done.

Competing interests: If there are none to declare, please write only "The authors declare that they have no competing interests". Done.

Tables: Please move your tables to after the reference list. Done.

Yours sincerely,
on behalf of all authors,

Jeannette Hofmeijer, MD

J. Hofmeijer, MD
Department of Neurology (C03.236)
University Medical Centre Utrecht
PO Box 85500
3508 GA Utrecht
the Netherlands
e-mail: j.hofmeijer@umcutrecht.nl
Telephone: 0031-30-2508600